Author's response to reviews

Title: Patient adherence to and tolerability of self-administered interferon beta-1a using an electronic autoinjection device: a multicentre, open-label, phase IV study

Authors:

Alessandra Lugaresi (a.lugaresi@unich.it)
Ciro Florio (mamifaccilpiacere@tin.it)
Vincenzo Brescia-Morra (centrosm@unina.it)
Salvatore Cottone (salvatore.cottone61@libero.it)
Paolo Bellantonio (pbellantonio@yahoo.it)
Marinella Clerico (marinella.clerico@unito.it)
Diego Centonze (centonze@uniroma2.it)
Antonio Uccelli (auccelli@neurologia.unige.it)
Maria di Ioia (maria.diioia@unich.it)
Giovanna De Luca (gio_de Luca@yahoo.it)
Andrea Marcellusi (andrea.marcellusi@uniroma2.it)
Andrea Paolillo (andrea.paolillo@merckgroup.com)
the BRIDGE study group (alessandra.lugaresi@virgilio.it)

Version: 2 Date: 14 October 2011

Author's response to reviews: see over
Re: Patient adherence to and tolerability of self-administered interferon β-1a using an electronic autoinjection device: a multicentre, open-label, phase IV study

Dear Editors:

We have provided the registration number, as requested below.

Thank you kindly.

MS: 1125498122614170
Research article
Patient adherence to and tolerability of self-administered interferon beta-1a using an electronic autoinjection device: a multicentre, open-label, phase IV study
Alessandra Lugaresi, Ciro Florio, Vincenzo Brescia-Morra, Salvatore Cottone, Paolo Bellantonio, Marinella Clerico, Diego Centonze, Antonio Uccelli, Maria di Ioia, Giovanna De Luca, Andrea Marcellusi, Andrea Paolillo and the BRIDGE study group
BMC Neurology

Dear Prof Alessandra Lugaresi,

Thank you for your recent submission of the manuscript detailed above. Before we can proceed with peer review we will need you to make some changes to the manuscript. We strongly urge you to make these changes promptly, as we cannot start the peer review process until we have received a version containing the changes.

1. Requesting trial registration number: We note that you report a EudraCT number, please note that we do not accept trial registration at this registry as it is not publicly available. Please could you provide us with a trial registration number from a registry that we accept.

BioMed Central has always supported initiatives to improve the performance and reporting of clinical trials, part of which includes prospective registering and numbering of trials. BioMed Central requests a trial registration number for manuscripts reporting work that falls within the International Committee of Medical Journal Editors (ICMJE) definition of a clinical trial: any research study that prospectively assigns human subjects to one or more health related interventions to evaluate the effects on health outcomes.

We would like you to confirm that your clinical trial is in a publicly accessible registry before we begin peer review. The trial registration number should be included as the last line of the abstract of the manuscript.

Please note that we only accept registration numbers issued by registries that meet all of the ICMJE criteria (http://www.icmje.org/publishing_10register.html). Registries which meet the requirements of the ICMJE include WHO Primary Registries (http://www.who.int/ictrp/network/primary/en/index.html).

Once you know your trial registration number, please submit a revised version of your manuscript with the number included in the abstract. The last section of the abstract should be Trial Registration: listing the trial registration number.

Thank you for your cooperation.
registry and the unique identifying number, e.g. Trial registration: Current Controlled Trials ISRCTN73824458. Please note that there should be no space between the letters and numbers of the trial registration number.

If you have applied to register your trial but are experiencing delays please provide us with the details of your registration.

Once you have made these changes, please upload the revised version by following these instructions:

1. Go to 'my BioMed Central' (http://www.biomedcentral.com/my/) and log on with your email address and password. Then click on 'My manuscripts'. You will reach a page giving details of all the manuscripts you have submitted.
2. Click on the 'Submit revision' button next to the title of this manuscript.
3. In the 'Manuscript details' tab, please update the title, abstract and author details if they have changed since the previous version.
4. In the 'Cover letter' tab, please provide a covering letter with a point-by-point description of the changes made.
5. In the 'Upload files' tab, please upload the revised version of the manuscript and press 'Submit new version'. Please wait for the confirmation page to appear - this may take a few moments.

We would be grateful if you could resubmit your files with the changes requested within the next two weeks. If we haven't heard from you or received the updated version by that time I'm afraid we may need to close the file, as we cannot keep submissions on hold indefinitely.

Please do not hesitate to contact us if you have any questions.

Best wishes,

Arnold Bongcayao
Journal Editorial Office
BioMed Central

Web: biomedcentral.com

Sincerely,

Alessandra Lugaresi
Department of Neuroscience and Imaging
Università degli Studi “Gabriele d’Annunzio”
Via dei Vestini 31
66100 Chieti
Italy
Tel: +39 (0871) 358532
Fax +39 0871 358523
Email: a.lugaresi@unich.it; alessandra.lugaresi@virgilio.it